

1

2

3

5

6 7 **ARTICLE IN PRESS**





Speech Communication xxx (2010) xxx-xxx

www.elsevier.com/locate/specom

Automatic speech emotion recognition using modulation spectral features

Siqing Wu^{a,*}, Tiago H. Falk^b, Wai-Yip Chan^a

^a Department of Electrical and Computer Engineering, Queen's University, Kingston, ON, Canada K7L 3N6

^b Institute of Biomaterials and Biomedical Engineering, University of Toronto, Toronto, ON, Canada M5S 3G9

8 Abstract

9 In this study, modulation spectral features (MSFs) are proposed for the automatic recognition of human affective information from speech. The features are extracted from an auditory-inspired long-term spectro-temporal representation. Obtained using an auditory filt-10 erbank and a modulation filterbank for speech analysis, the representation captures both acoustic frequency and temporal modulation 11 12 frequency components, thereby conveying information that is important for human speech perception but missing from conventional 13 short-term spectral features. On an experiment assessing classification of discrete emotion categories, the MSFs show promising perfor-14 mance in comparison with features that are based on mel-frequency cepstral coefficients and perceptual linear prediction coefficients, two 15 commonly used short-term spectral representations. The MSFs further render a substantial improvement in recognition performance 16 when used to augment prosodic features, which have been extensively used for emotion recognition. Using both types of features, an overall recognition rate of 91.6% is obtained for classifying seven emotion categories. Moreover, in an experiment assessing recognition 17 18 of continuous emotions, the proposed features in combination with prosodic features attain estimation performance comparable to 19 Q1 human evaluation.

20 Q2 © 2010 Elsevier B.V. All rights reserved.

21 Keywords: Emotion recognition; Speech modulation; Spectro-temporal representation; Affective computing; Speech analysis 22

1. Introduction 23

Affective computing, an active interdisciplinary research 24 field, is concerned with the automatic recognition, interpre-25 26 tation, and synthesis of human emotions (Picard, 1997). Within its areas of interest, speech emotion recognition 27 (SER) aims at recognizing the underlying emotional state 28 of a speaker from the speech signal. The paralinguistic 29 information conveyed by speech emotions has been found 30 to be useful in multiple ways in speech processing, espe-31 cially serving as an important ingredient of "emotional 32 intelligence" of machines and contributing to human-33 34 machine interaction (Cowie et al., 2001; Ververidis and Kotropoulos, 2006). Moreover, since a broad range of 35 emotions can be faithfully delivered in a telephone conver-36 sation where only auditory information is exchanged, it 37 should be possible to build high-performance emotion rec-38 ognition systems, using only speech signals as the input. 39 Such speech based systems can function either indepen-40 dently or as modules of more sophisticated techniques that 41 combine other information sources such as facial expres-42 sion and gesture (Gunes and Piccard, 2007). 43

Despite the substantial advances made in this area, SER 44 still faces a number of challenges, one of which is designing 45 effective features. Most acoustic features that have been 46 used for emotion recognition can be divided into two cate-47 gories: prosodic and spectral. Prosodic features have been 48 shown to deliver important emotional cues of the speaker 49 (Cowie et al., 2001; Ververidis and Kotropoulos, 2006; 50 Busso et al., 2009). Even though there is no agreement 51 on the best features to use, prosodic features form the most 52

Corresponding author.

E-mail addresses: siging.wu@queensu.ca (S. Wu), chan@queensu.ca (W.-Y. Chan).

^{0167-6393/\$ -} see front matter © 2010 Elsevier B.V. All rights reserved. doi:10.1016/j.specom.2010.08.013

127

128

129

130

131

132

133

134

135

136

137

7 September 2010 Disk Used

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

commonly used feature type for SER, and have been 53 extensively studied by previous works (e.g. Cowie and 54 Douglas-Cowie, 1996; Abelin and Allwood, 2000; Cowie 55 et al., 2001; Mozziconacci, 2002; Scherer, 2003; Barra 56 et al., 2006; Schuller et al., 2007a; Busso et al., 2009). On 57 the other hand, spectral features (including cepstral fea-58 59 tures) also play a significant role in SER as they convey the frequency content of the speech signal, and provide 60 complementary information to prosodic features. Compar-61 atively, however, limited research efforts have been put into 62 constructing more powerful spectral features for emotion 63 recognition. The spectral features are usually extracted 64 over a short frame duration (e.g. 20-30 ms), with longer 65 temporal information incorporated in the form of local 66 derivatives (e.g. Nwe et al., 2003; Batliner et al., 2006; Vla-67 senko et al., 2007). 68

The limitations of short-term spectral features for 69 speech recognition, however, are considerable (Morgan 70 et al., 2005). Even with the inclusion of local derivatives. 71 the fundamental character of the features remains fairly 72 73 short-term. In short, conventional spectral features used 74 for speech recognition, such as the well-known mel-fre-75 quency cepstral coefficients (MFCCs), convey the signal's short-term spectral properties only, omitting important 76 temporal behavior information. Such limitations are also 77 likely to hamper SER performance. On the other hand, 78 advances in neuroscience suggest the existence of spectro-79 80 temporal (ST) receptive fields in mammalian auditory cortex which can extend up to temporal spans of hundreds of 81 82 milliseconds and respond to modulations in the time-frequency domain (Depireux et al., 2001; Shamma, 2001; 83 Chih et al., 2005). The importance of the modulation spec-84 trum of speech is evident in a number of areas, including 85 auditory physiology, psychoacoustics, speech perception, 86 and signal analysis and synthesis, as summarized in (Atlas 87 and Shamma, 2003). These new insights further reveal the 88 shortcomings of short-term spectral features as they dis-89 90 card the long-term temporal cues used by human listeners, 91 and highlight the need for more perceptually motivated features. 92

In line with these findings, long-term modulation spec-93 tral features (MSFs) are proposed in this paper for emotion 94 recognition. These features are based on frequency analysis 95 96 of the temporal envelopes (amplitude modulations) of multiple acoustic frequency bins, thus capturing both spectral 97 and temporal properties of the speech signal. The proposed 98 features are applied to two different SER tasks: (1) classifi-99 cation of discrete emotions (e.g. joy, neutral) under the cat-100 101 egorical framework which characterizes speech emotions using categorical descriptors and (2) estimation of continu-102 ous emotions (e.g. valence, activation) under the dimen-103 sional framework which describes speech emotions as 104 points in an emotion space. In the past, classification tasks 105 106 have drawn dominant attention of the research community (Cowie et al., 2001; Douglas-Cowie et al., 2003; Ververidis 107 and Kotropoulos, 2006; Shami and Verhelst, 2007). Recent 108 studies, however, have also focused on recognizing contin-109

uous emotions (Grimm et al., 2007a,b; Wollmer et al., 110 2008; Giannakopoulos et al., 2009). 111

To our knowledge, the only previous attempt at using 112 modulation spectral content for the purpose of emotion 113 recognition is reported in (Scherer et al., 2007), where the 114 modulation features are combined with several other fea-115 ture types (e.g., loudness features) and approximately 116 70% recognition rate is achieved on the so-called Berlin 117 emotional speech database (Burkhardt et al., 2005). This 118 present study, which extends our previous work (Wu 119 et al., 2009), is different in several ways, namely (1) filter-120 banks are employed for spectral decomposition; (2) the 121 proposed MSFs are designed by exploiting a long-term 122 ST representation of speech, and are shown to achieve con-123 siderably better performance on the Berlin database rela-124 tive to (Scherer et al., 2007); and (3) continuous emotion 125 estimation is also performed. 126

The remainder of the paper is organized as follows. Section 2 presents the algorithm for generating the long-term ST representation of speech. Section 3 details the MSFs proposed in this work, as well as short-term spectral features and prosodic features extracted for comparison purposes. Section 4 introduces the databases employed. Experimental results are presented and discussed in Section 5, where both discrete emotion classification (Section 5.1) and continuous emotion estimation (Section 5.2) are performed. Finally, Section 6 gives concluding remarks.

2. ST representation of speech

The auditory-inspired spectro-temporal (ST) representa-138 tion of speech is obtained via the steps depicted in Fig. 1. 139 The initial pre-processing module resamples the speech sig-140 nal to 8 kHz and normalizes its active speech level to 141 -26 dBov using the P.56 speech voltmeter (Intl. Telecom. 142 Union, 1993). Since emotions can be reliably conveyed 143 through band-limited telephone speech, we consider the 144 8 kHz sampling rate adequate for SER. Speech frames 145 (without overlap) are labeled as active or inactive by the 146 G.729 voice activity detection (VAD) algorithm described 147 in (Intl. Telecom. Union, 1996) and only active speech 148 frames are retained. The preprocessed speech signal s(n)149 is framed into long-term segments $s_k(n)$ by multiplying a 150 256 ms Hamming window with 64 ms frame shift, where 151 k denotes the frame index. Because the first subband filter 152 in the modulation filterbank (described below) analyzes 153 frequency content around 4 Hz, this relatively long tempo-154 ral span is necessary for such low modulation frequencies. 155





ARTICLE IN PRESS

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

156 It is well-known that the human auditory system can be 157 modeled as a series of over-lapping band-pass frequency 158 channels (Fletcher, 1940), namely auditory filters with crit-159 ical bandwidths that increase with filter center frequencies. 160 The output signal of the *i*th critical-band filter at frame k is 161 given by:

164
$$s_k(i,n) = s_k(n) * h(i,n),$$
 (1)

where h(i, n) denotes the impulse response of the *i*th chan-165 nel, and * denotes convolution. Here, a critical-band 166 gammatone filterbank (Aertsen and Johannesma, 1980) 167 with N subband filters is employed. The implementation 168 in (Slaney, 1993) is used. The center frequencies of these fil-169 ters (namely *acoustic* frequency, to distinguish from *modu*-170 lation frequency of the modulation filterbank) are 171 172 proportional to their bandwidths, which in turn, are char-173 acterized by the equivalent rectangular bandwidth (Glasberg and Moore, 1990): 174

$$ERB_i = \frac{F_i}{Q_{ear}} + B_{min}, \qquad (2)$$

where F_i is the center frequency (in Hz) of the *i*th critical-178 band filter, and Q_{ear} and B_{min} are constants set to 179 9.26449 and 24.7, respectively. In our simulations, a gamm-180 atone filterbank with 19 filters is used, where the first and 181 the last filters are centered at 125 Hz and 3.5 kHz, with 182 183 bandwidths of 38 and 400 Hz, respectively. The magnitude response of the filterbank is depicted in Fig. 2. The 184 temporal envelope, or more specifically, the Hilbert enve-185 lope $\mathcal{H}_k(i, n)$, is then computed from $s_k(i, n)$ as the magni-186 tude of the complex analytic signal $\hat{s}_k(i,n) = s_k(i,n) + s_k(i,n)$ 187 188 $i\mathbb{H}\{s_k(i,n)\}$, where $\mathbb{H}\{\cdot\}$ denotes the Hilbert transform. Hence. 189

192
$$\mathcal{H}_k(i,n) = |\hat{s}_k(i,n)| = \sqrt{s_k^2(i,n) + \mathbb{H}^2\{s_k(i,n)\}}.$$
 (3)

Fig. 3 shows an example of a bandpassed speech segment (subplot a) and its Hilbert envelope (subplot b).

The auditory spectral decomposition modeled by the critical-band filterbank, however, only comprises the first stage of the signal transformation performed in the human



Fig. 2. Magnitude response of a 19-band auditory filterbank with center frequencies ranging from 125 Hz to 3.5 kHz.



Fig. 3. Example of Hilbert envelope: (a) a 125 ms output of a critical-band filter centered at 650 Hz and (b) the corresponding Hilbert envelope.

auditory system. The output of this early processing is 198 further interpreted by the auditory cortex to extract spec-199 tro-temporal modulation patterns (Shamma, 2003; Chih 200 et al., 2005). An *M*-band modulation filterbank is 201 employed in addition to the gammatone filterbank to 202 model such functionality of the auditory cortex. By apply-203 ing the modulation filterbank to each $\mathcal{H}_k(i, n)$, M outputs 204 $\mathcal{H}_k(i, j, n)$ are generated where *j* denotes the *j*th modulation 205 filter, $1 \leq j \leq M$. The filters in the modulation filterbank 206



Fig. 4. Magnitude response of a 5-band modulation filterbank with center frequencies ranging from 4 to 64 Hz.



Fig. 5. $E_k(i,j)$ for one frame of a "*neutral*" speech file: low channel index indicates low frequency.

Please cite this article in press as: Wu, S. et al., Automatic speech emotion recognition using modulation spectral features, Speech Comm. (2010), doi:10.1016/j.specom.2010.08.013

217

219



S. Wu et al. | Speech Communication xxx (2010) xxx-xxx



Fig. 6. Average E(i,j) for seven emotion categories; for each emotion, $E_k(i,j)$ is averaged over all frames from all speakers of that emotion; "AC" and "MC" denote the acoustic and modulation frequency channels, respectively.

207 are second-order bandpass with quality factor set to 2, as suggested in (Ewert and Dau, 2000). In this work we use 208 an M = 5 filterbank whose filter center frequencies are 209 equally spaced on logarithm scale from 4 to 64 Hz. The filt-210 erbank was shown in preliminary experiments to strike a 211 good balance between performance and model complexity. 212 The magnitude response of the modulation filterbank is 213 depicted in Fig. 4. 214

Lastly, the ST representation $E_k(i,j)$ of the *k*th frame is obtained by measuring the energy of $\mathcal{H}_k(i,j,n)$, given by:

$$E_k(i,j) = \sum_{n=1}^{L} |\mathcal{H}_k(i,j,n)|^2,$$
(4)

where $1 \le k \le T$ with *L* and *T* representing the number of 220 samples in one frame and the total number of frames, 221 respectively. For a fixed $j = j^*$, $E_k(i, j^*)$ relates the auditory 222 spectral samples of modulation channel j^* after critical-223 band grouping. An example of $E_k(i,j)$ is illustrated in 224 Fig. 5. By incorporating the auditory filterbank and the 225 modulation filterbank, a richer two-dimensional frequency 226 representation is produced and allows for analysis of mod-227 ulation frequency content across different acoustic fre-228 quency channels. 229

Fig. 6 shows the ST representation E(i, j) for the seven 230 emotions in the Berlin database (cf. Section 4.1), where 231 every E(i, j) shown is the average over all the frames and 232 speakers available in the database for an emotion. As illus-233 234 trated in the figure, the average ST energy distribution over the joint acoustic-modulation frequency plane is similar for 235 some emotions (e.g. anger vs. joy), suggesting they could 236 become confusion pairs, while very distinct for some others 237

(e.g. anger vs. sadness), suggesting they could be well dis-238 criminated from each other. As reasonably expected, the 239 less expressive emotions such as *boredom* and *sadness* have 240significantly more low acoustic frequency energy than 241 anger and joy (see also Fig. 7), corroborating findings in 242 previous studies (Cowie et al., 2001; Scherer, 2003). The 243 ST distribution for *neutral* peaks at 4 Hz modulation fre-244 quency, matching the nominal syllabic rate (Kanederaa 245 et al., 1999). The peak shifts to a higher modulation fre-246 quency for anger, joy, and fear, suggesting a faster speaking 247 rate for these emotions. Less expressive emotions such as 248 boredom and sadness exhibit more prominently lowpass 249 modulation spectral shapes, suggestive of lower speaking 250 rates. Interestingly, sadness also shows increased energy 251 for the last two modulation channels (centered at 32 and 252 64 Hz, respectively) relative to *anger* and *joy* (not evident 253 from the plots, as the dominant energy is concentrated in 254



Fig. 7. Estimated pdfs of $\overline{\Phi}_3(1)$ for three basic emotions.

ARTICLE IN PRESS

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

5

307

308

309

310

311

312

313

314

315

316

317

318

319

320

321

322

323

324

325

326

327

328

329

330

331

332

333

334

335

336

337

338

339

340

lower modulation channels and the absolute amount of
energy at higher modulation frequencies is small). This
might be due to the fact that sad speech is more breathy
(Ishi et al., 2010), a phenomenon somewhat analogous to
reverberant speech, whose effectively unvoiced excitation
engenders more high modulation frequency energy (Falk
and Chan, 2010b).

262 **3. Feature extraction**

In this section, we detail the proposed MSFs extracted from the ST representation. Short-term spectral features and prosodic features considered in our experiments are also described.

267 3.1. Modulation spectral features

Two types of MSFs are calculated from the ST represen-268 tation, by means of spectral measures and linear prediction 269 parameters. For each frame k, the ST representation $E_k(i, j)$ 270 271 is scaled to unit energy before further computation, i.e. 272 $\sum_{i,j} E_k(i,j) = 1$. Six spectral measures $\Phi_1 - \Phi_6$ are then cal-273 culated on a per-frame basis. For frame k, $\Phi_{1,k}(j)$ is defined 274 as the *mean* of the energy samples belonging to the *j*th modulation channel $(1 \le i \le 5)$: $275 \\ 276$

278
$$\Phi_{1,k}(j) = \frac{\sum_{i=1}^{N} E_k(i,j)}{N}.$$
 (5)

Parameter Φ_1 characterizes the energy distribution of speech along the modulation frequency. The second spectral measure is the *spectral flatness* which is defined as the ratio of the geometric mean of a spectral energy measure to the arithmetic mean. In our calculation, $E_k(i,j)$ is used as the spectral energy measure at frame k for modulation band j and Φ_2 is thus defined as:

$$\Phi_{2,k}(j) = \frac{\sqrt[N]{\prod_{i=1}^{N} E_k(i,j)}}{\Phi_{1,k}(j)}.$$
(6)

A spectral flatness value close to 1 indicates a flat spectrum, while a value close to 0 suggests a spectrum with widely different spectral amplitudes. The third measure employed is the *spectral centroid* which provides a measure of the "center of mass" of the spectrum in each modulation channel. Parameter Φ_3 for the *j*th modulation channel is computed as:

$$\Phi_{3,k}(j) = \frac{\sum_{i=1}^{N} f(i) E_k(i,j)}{\sum_{i=1}^{N} E_k(i,j)}.$$
(7)

298

Two types of frequency measure f(i) have been experi-299 mented: (1) f(i) being the center frequency (in Hz) of the 300 *i*th critical-band filter of the auditory filterbank and (2) 301 302 f(i) being the index of the *i*th critical band filter, i.e., f(i) =303 *i*. No remarkable difference in performance is observed be-304 tween the two measures, thus the latter is chosen for simplicity. Moreover, given the observation that adjacent 305 modulation channels usually have considerable correlation, 306

the spectral flatness and the centroid parameters of adjacent modulation channels also exhibit high correlation. In order to alleviate such information redundancy, $\Phi_{2,k}(j)$ and $\Phi_{3,k}(j)$ are only computed for $j \in \{1,3,5\}$.

Among the three aforementioned spectral measures, Φ_3 is observed to be particularly useful. Fig. 7 illustrates a representative example, where $\overline{\Phi}_3(1)$ is the average of $\Phi_{3k}(1)$ computed over each utterance in the Berlin database belonging to three basic emotions: anger, neutral, and sadness, and the probability density function (PDF) of the averages for each emotion is estimated as a unimodal Gaussian. Considering *neutral* as a reference point, *anger* and sadness display an upward and downward shift of spectral centroid in acoustic frequency, respectively. This result is consistent with the ST patterns of these three emotions displayed in Fig. 6. Even though the PDFs of sadness and *neutral* overlap to some extent, good separation is shown for anger vs. neutral, and almost perfect discrimination is achieved between anger and sadness, using only one feature.

In addition to parameters that measure the spectral behavior of each individual modulation channel, additional spectral measures that measure the relationship of different modulation channels are computed. First, the 19 acoustic channels are grouped into four divisions: 1–4, 5–10, 11–15, and 16–19, namely D_l ($1 \le l \le 4$), which roughly correspond to frequency regions of <300, 300–1000, 1000–2000, and >2000 Hz, respectively, and have been shown in pilot experiment to achieve a good compromise between the amount of fine details extracted from data and performance. Channels in the same division are summed: $\mathbb{E}_k(l,j) = \sum_{i \in D_l} E_k(i,j)$. Then the *modulation spectral centroid* (Φ_4) is calculated in a manner similar to Eq. 7:

$$\Phi_{4,k}(l) = \frac{\sum_{j=1}^{M} j \mathbb{E}_k(l,j)}{\sum_{j=1}^{M} \mathbb{E}_k(l,j)}.$$
(8)
342

Unlike $\Phi_{3,k}(j)$ which measures the spectral centroid in the 343 acoustic frequency domain for modulation band j, $\Phi_{4,k}(l)$ 344 calculates the centroid in the modulation frequency domain 345 for D_l . The last two spectral measures $\Phi_{5,k}(l)$ and $\Phi_{6,k}(l)$ are 346 the linear regression coefficient (slope) and the correspond-347 ing regression error (root mean squared error, RMSE) 348 obtained by fitting a first-degree polynomial to $\mathbb{E}_k(l, j)$, 349 $j = 1, \ldots, M$, in a least squares sense. By calculating Φ_{4-} 350 Φ_6 , information is extracted about the rate of change of 351 the selected acoustic frequency regions, thereby compactly 352 capturing the temporal dynamic cues. In total, 23 features 353 are obtained from the ST representation per frame by 354 applying the six spectral measures. 355

Besides taking the spectral measures described above, 356 linear predication (LP) analysis is further applied to 357 selected modulation channels *j* where $j \in \{1,3,5\}$, to extract 358 the second set of MSFs from $E_k(i,j)$. This selection of 359 modulation channels is also for the purpose of reducing 360 information redundancy caused by high correlation 361 between adjacent channels. The autocorrelation method 362

421

435

7 September 2010 Disk Used

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

for autoregressive (AR) modeling is used here. In order to 363 suppress local details while preserving the broad structure 364 beneficial to recognition, a 5th-order all-pole model is used 365 to approximate the spectral samples. The computational 366 367 cost of this AR modeling is negligible due to the low LP order and the small number of spectral samples per modu-368 369 lation channel (19 here). The LP coefficients obtained are further transformed into cepstral coefficients (LPCCs), 370 and denoted as $C_k(n, j)$ $(0 \le n \le 5)$. The LPCCs have been 371 shown to be generally more robust and reliable for speech 372 recognition than the direct LP coefficients (Rabiner and 373 Juang, 1993). We have tested both types and indeed the 374 LPCCs yield better recognition performance in our appli-375 cation. Together with the 23 aforementioned features, a 376 total of 41 MSFs are calculated frame-by-frame. 377

Although MSFs are extracted at a frame-level (FL), the 378 common approach in current SER literature computes fea-379 tures at the utterance-level (UL) (e.g. Grimm et al., 380 2007a,b; Shami and Verhelst, 2007; Schuller et al., 2007b; 381 Clavel et al., 2008; Lugger and Yang, 2008; Busso et al., 382 2009; Giannakopoulos et al., 2009; Sun and Torres, 383 384 2009), by applying descriptive functions (typically statistical) to the trajectories of FL features (and often also their 385 derivatives to convey local dynamic information). The UL 386 features capture the global properties and behaviors of 387 388 their FL counterparts. It is desirable for the UL features to capture the supra-segmental characteristics of emotional 389 speech that can be attributed to emotions rather than spe-390 cific spoken-content and to effectively avoid problems such 391 as spoken-content over-modeling (Vlasenko et al., 2007). 392 Following the UL approach, two basic statistics: mean 393 and standard deviation (std. dev.) of the FL MSFs are cal-394 culated in this work, producing 82 UL MSFs. 395

396 3.2. Short-term spectral features

397 3.2.1. MFCC features

398 The mel-frequency cepstral coefficients (MFCCs), first introduced in (Davis and Mermelstein, 1980) and success-399 fully applied to automatic speech recognition, are popular 400 short-term spectral features used for emotion recognition. 401 They are extracted here for comparison with the proposed 402 403 long-term MSFs. The speech signal is first filtered by a 404 high-pass filter with a pre-emphasis coefficient of 0.97, and the first 13 MFCCs (including the zeroth order log-405 energy coefficient) are extracted from 25 ms Hamming-win-406 dowed speech frames every 10 ms. As a common practice, 407 the delta and double-delta MFCCs describing local dynam-408 409 ics are calculated as well to form a 39-dimensional FL feature vector. The most frequently used UL MFCC features 410 411 for emotion recognition include mean and std. dev. (or variance) of the first 13 MFCCs and their deltas (e.g. Grimm 412 et al., 2007a,b; Schuller et al., 2007a; Vlasenko et al., 2007; 413 414 Clavel et al., 2008; Lugger and Yang, 2008). In this work, we compute mean, std. dev., and 3rd-5th central moments 415 of the first 13 MFCCs, as well as their deltas and double-416 deltas, giving 195 MFCC features in total. This MFCC 417

feature set is an extension to the one used in (Grimm 418 et al., 2007a, 2008), by further considering the delta 419 coefficients. 420

3.2.2. PLP features

In addition to MFCCs, perceptual linear predictive 422 (PLP) coefficients (Hermansky, 1990) are also extracted 423 from speech, serving as an alternative choice of short-term 424 spectral features for comparison. PLP analysis approxi-425 mates the auditory spectrum of speech by an all-pole model 426 that is more consistent with human hearing than conven-427 tional linear predictive analysis. A 5th-order model is 428 employed as suggested in (Hermansky, 1990). The PLP 429 coefficients are transformed to cepstral coefficients c(n)430 $(0 \le n \le 5)$. The delta and double-delta coefficients are also 431 considered. The aforementioned statistical parameters as 432 used for MFCC are calculated for the PLP coefficients, giv-433 ing 90 candidate PLP features. 434

3.3. Prosodic features

Prosodic features have been, among numerous acoustic 436 features employed for SER, the most widely used feature 437 type as mentioned in Section 1. Hence they are used here 438 as a benchmark, and more importantly, to verify whether 439 the MSFs can serve as useful additions to the extensively 440 used prosodic features. The most commonly used prosodic 441 features are based on pitch, intensity, and speaking rate. 442 The features are estimated on a short-term frame basis. 443 and their contours are used to compute UL features. The 444 statistics of these trajectories are shown to be of fundamen-445 tal importance for conveying emotional cues (Cowie et al., 446 2001; Nwe et al., 2003; Ververidis and Kotropoulos, 2006; 447 Busso et al., 2009). 448

In total, 75 prosodic features are extracted as listed in Table 1. Note that a complete coverage of prosodic features is infeasible. Consequently, the features calculated here are by no means exhaustive, but serve as a representative sampling of the essential prosodic feature space. Pitch is computed for voiced speech using the pitch tracking algorithm in (Talkin, 1995), and intensity is measured for

Table 1

List of prosodic features.	
Pitch, intensity, delta-pitch, delta-intensity Mean, std. dev., skewness, kurtosis, shimmer, maximum, minimum, median, quartiles, range, differences between quartiles, linear & quadratic regression coefficients, regression error (RMSE)	
Speaking rate Mean and std. dev. of syllable durations Ratio between the duration of voiced and unvoiced speech	

Others Zero-crossing rate (ZCR) Teager energy operator (TEO)

7

508

509

510

511

512

513

514

515

516

517

518

519

520

521

522

523

524

525

526

548

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

456 active speech in dB. Note that shimmer is computed for the 457 trajectories of pitch and intensity only, not their deltas. For 458 a sequence x_n of length N_x , shimmer in this work is calcu-459 lated as:

462
$$S = \frac{\frac{1}{N_x - 1} \sum_{n=1}^{N_x - 1} |x_n - x_{n+1}|}{\frac{1}{N_x} \sum_{n=1}^{N_x} x_n}.$$
 (9)

463 Features related to speaking rate are also extracted using syllabic and voicing information as shown in the table. 464 465 Moreover, the zero-crossing rate (ZCR) and the Teager energy operator (TEO) (Kaiser, 1990) of the speech signal are 466 calculated as well, though they do not directly relate to 467 prosody. TEO extracts useful information about the non-468 linear airflow structure of speech production (Zhou et al., 469 2001). The TEO for a discrete-time signal x_n is defined as: 470 471

473
$$\operatorname{TEO}(x_n) = x_n^2 - x_{n+1}x_{n-1}.$$
 (10)

The mean Teager energy of the speech signal is used as a feature, and is found to be an effective feature in our experiments (cf. Section 5.1.3).

477 4. Emotional speech data

478 *4.1. Berlin emotional speech database*

The Berlin emotional speech database (Burkhardt et al., 479 2005) is used for experiments classifying discrete emotions. 480 This publicly available database is one of the most popular 481 databases used for emotion recognition, thus facilitating 482 comparisons with other works. Ten actors 5m/5f) each 483 484 uttered 10 everyday sentences (five short and five long, typically between 1.5 and 4 s) in German, sentences that can 485 be interpreted in all of seven emotions acted. The raw data-486 base (prior to screening) has approximately 800 sentences 487 and is further evaluated by a subjective perception test with 488 20 listeners. Utterances scoring higher than 80% emotion 489 490 recognition rate and considered natural by more than 60% listeners are included in the final database. The num-491 bers of speech files for the seven emotion categories in the 492 screened Berlin database are: anger (127), boredom (81), 493 disgust (46), fear (69), joy (71), neutral (79) and sadness 494 (62). 495

496 4.2. Vera am Mittag (VAM) database

The VAM database (Grimm et al., 2008) is a relatively 497 new database containing spontaneous emotions, created 498 within a three-dimensional emotion space framework. It 499 was recorded from a German TV talk-show "Vera am Mit-500 tag". There are three individual modules in the complete 501 VAM database: VAM-Audio, VAM-Video, and VAM-502 503 Faces, containing audio signal only, audio + visual signals, 504 and face images, respectively. In this work, only the VAM-505 Audio module is employed and hereinafter it is simply referred to as the VAM database. The recordings are man-506 507 ually segmented at the utterance level. Emotions in the



Fig. 8. Distribution of the emotion primitives in the overall VAM database (Grimm et al., 2008).

database are described in an emotion space consisting of three emotion primitives: *valence* (or *evaluation*, ranging from negative to positive), activation (with levels from low to high) and *dominance* (the apparent strength of the speaker, i.e. ability to handle a situation) (Grimm et al., 2007a). The continuous emotion values on each primitive scale are obtained through subjective assessment. The VAM database contains two parts: VAM I with 478 utterances from 19 speakers (4m/15f) and 17 human evaluators assessing the primitives, and VAM II with 469 utterances from 28 speakers (7m/21f) and six evaluators. The distributions of the three primitives in the VAM database (I + II)are shown in Fig. 8 (Grimm et al., 2008). It can be seen from the histograms that compared to activation and dominance, the distribution of valence is less balanced. The database contains mostly neutral and negative emotions, due to the topics discussed in the talk-show (Grimm et al., 2008).

5. Experiments

In this section, results of the experimental evaluation are 527 presented. Support vector machines (SVMs) (Vapnik, 528 1995) are used for recognition of both discrete and contin-529 uous emotions. While support vector classification finds the 530 separation hyperplane that maximizes the margin between 531 two classes, support vector regression determines the 532 regression hyperplane that approximates most data points 533 with ϵ precision. The SVM implementation in (Chang and 534 Lin, 2009) is adopted with the radial basis function (RBF) 535 kernel employed. The design parameters of SVM are 536 selected using training data via a grid search on a base-2 537 logarithmic scale. In general, the RBF kernel can be a good 538 choice as justified in (Hsu et al., 2007) because: (1) it can 539 model the non-linear relation between attributes and target 540 values well; (2) the linear kernel is a special case of RBF 541 kernel; (3) it has less hyperparameters than the polynomial 542 kernel; and (4) it has less numerical difficulties compared to 543 polynomial and sigmoid kernels. Features from training 544 data are linearly scaled to [-1,1] before applying SVM, 545 with features from test data scaled using the trained linear 546 mapping function as suggested in (Hsu et al., 2007). 547

5.1. Experiment I: discrete emotion classification

All results achieved on the Berlin database are produced 549 using 10-fold cross-validation. The data partitioning is 550

8

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

based on random sampling of files from a pool wherein all 551 speakers are mixed; hence the cross-validation is not 552 speaker-independent. Moreover, samples in each class have 553 been randomly divided into 10 disjoint subsets approxi-554 555 mately equal in size. Each validation trial takes nine subsets from every class for training, with the remaining 556 557 subset kept unseen from the training phase and used for testing only. 558

A two-stage feature selection scheme is employed and is 559 described in Section 5.1.1. The proposed MSFs are initially 560 compared to MFCC and PLP features in Section 5.1.2. 561 with their contribution as supplementary features to pro-562 sodic features investigated in Section 5.1.3. The effect of 563 taking a speaker normalization (SN) step to pre-process 564 the data prior to data partitioning for cross-validation is 565 also investigated, where features are first mean and vari-566 ance normalized within the scope of each speaker to com-567 pensate for speaker variations, as performed in (Vlasenko 568 et al., 2007). Let $f_{u,v}(n)$ $(1 \le n \le N_{u,v})$ stand for the *u*th fea-569 ture from speaker v with $N_{u,v}$ denoting its sample size which 570 in our case, is the number of all available samples in the 571 572 database from that speaker. Then the new feature $f'_{uv}(n)$ processed by SN is given by: 573 574

$$f'_{u,v}(n) = \frac{f_{u,v}(n) - \overline{f_{u,v}}}{\sqrt{\frac{1}{N_{u,v} - 1} \sum_{m=1}^{N_{u,v}} (f_{u,v}(m) - \overline{f_{u,v}})^2}},$$
(11)

where $\overline{f_{u,v}} = \frac{1}{N_{u,v}} \sum_{n=1}^{N_{u,v}} f_{u,v}(n)$. Since SN requires the system to know the speaker in advance, it might not be realistic in many applications. Nevertheless its results could still be interesting to see, hence included in following experiments.

5.1.1. Feature selection

576

582

593

Using all the features for machine learning might deteri-583 orate recognition performance due to the curse of dimen-584 585 sionality (Bishop, 2006). To this end, a two-stage feature selection scheme is proposed to reduce the number of fea-586 tures. The first stage calculates the Fisher discriminant 587 ratio (FDR) to rank each feature individually, which can 588 quickly eliminate irrelevant ("noisy") features. The normal-589 ized multi-class FDR for the *u*th feature is defined as: 590 591

FDR(u) =
$$\frac{2}{C(C-1)} \sum_{c_1} \sum_{c_2} \frac{(\mu_{c_1,u} - \mu_{c_2,u})^2}{\sigma_{c_1,u}^2 + \sigma_{c_2,u}^2},$$
 (12)

with $1 \leq c_1 \leq c_2 \leq C$, where $\mu_{c_1,u}$ and $\sigma_{c_1,u}^2$ are mean and 594 variance of the *u*th feature for the c_1 th class, and *C* is the 595 total number of classes. This FDR measure is normalized 596 597 by the number of binary comparisons made between two classes. The measure favors features with well-separated 598 means across classes and small within-class variances. Fea-599 tures of little discrimination power can then be removed by 600 601 FDR thresholding. Here the threshold is empirically set to 0.15 as further increasing the threshold results in no 602 improvement of performance, which reduces roughly 10% 603 of the proposed features and 15% of prosodic features, 604

but up to 50% of MFCC features and 40% of PLP features.605Thus the FDR step is important for the short-term spectral606feature pools, which would otherwise be rather noisy for607feature mining.608

In the second stage, two techniques are experimented to 609 obtain good features from the pre-screened feature pools 610 for SVM classification. The first technique is sequential 611 forward feature selection (SFS) (Kittler, 1978), which 612 iteratively augments the selected feature subset and consid-613 ers the combined effect of features and SVM classifier 614 during the evaluation process. The SFS algorithm also 615 helps to visualize changes of recognition accuracy as the 616 selected feature subset evolves, and thus provides a 617 straightforward way to compare performance of different 618 feature combinations. 619

The second technique is the well-known multi-class lin-620 ear discriminant analysis (LDA) (Bishop, 2006). It finds the 621 transformation optimizing the Fisher objective, that in turn 622 maximizes the between-class distance and minimizes the 623 within-class distance simultaneously. LDA can offer a dras-624 tic reduction in feature dimensionality for high-dimen-625 sional data and effectively alleviates the curse of 626 dimensionality, hence particularly useful in practice if the 627 size of the training data is limited relative to the number 628 of features. The main limitation of LDA is that it cannot 629 be applied to regression problems. LDA solves the general-630 ized eigen-problem: 631 632

$$\mathbf{S}_b \mathbf{w} = \lambda \mathbf{S}_w \mathbf{w},\tag{13}$$

where S_{h} and S_{w} are the between-class and within-class 635 scatter matrices, respectively. The eigenvectors \mathbf{w} are used 636 to form the columns of the transformation matrix W which 637 transforms data point x to $y = W^T x$. The components of y 638 constitute the LDA-transformed features. Since the maxi-639 mum rank of S_b for a C-class problem is C-1, the maxi-640 mum number of LDA features is also C - 1. For our 641 seven-class case, all six LDA-transformed features are used 642 to design SVM classifiers. 643



Fig. 9. Average FDR curves for the proposed, MFCC and PLP features (Berlin).

ARTICLE IN PRESS

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

Test	Feature	Method	SN	Recognition rate (%)							Average
				Anger	Boredom	Disgust	Fear	Joy	Neutral	Sadness	
#1	MSF (31/74)	SFS	No	92.1	86.4	73.9	62.3	49.3	83.5	98.4	79.6
	MFCC (49/92)			91.3	80.3	67.4	65.2	50.7	79.8	85.5	76.5
	PLP (48/51)			89.8	79.0	50.0	58.0	43.7	72.2	83.9	71.2
#2	MSF	LDA	No	91.3	86.4	78.3	71.0	60.6	83.5	88.7	81.3
	MFCC			83.5	85.2	78.3	76.8	53.5	79.8	83.9	77.9
	PLP			88.2	74.1	56.5	55.1	49.3	77.2	80.7	71.4
#3	MSF (41/73)	SFS	Yes	89.8	88.9	67.4	81.2	59.2	84.8	98.4	82.8
	MFCC (37/96)			88.2	82.7	71.7	76.8	54.9	76.0	90.3	78.5
	PLP (26/51)			90.6	72.8	56.5	66.7	46.5	81.0	95.2	75.1
#4	MSF	LDA	Yes	90.6	87.7	76.1	91.3	70.4	84.8	91.9	85.6
	MFCC			85.0	79.0	78.3	81.2	54.9	79.8	88.7	78.7
	PLP			90.6	80.3	58.7	72.5	56.3	81.0	88.7	77.8

Table 2 Recognition results for MSF, MFCC, and PLP features (Berlin): boldface indicates the best performance in each

644 5.1.2. Comparison with short-term spectral features

The proposed features are first compared to two short-645 term spectral feature types: MFCC and PLP features, by 646 means of FDR scores before applied to SVMs. The fea-647 648 tures are ranked by their FDR values (calculated using 649 all instances in the Berlin database before data partitioning). Curves depicting the FDR values averaged over the 650 top N_{fdr} FDR-ranked features are shown in Fig. 9 as a 651 function of N_{fdr} . These average FDR curves can be viewed 652 as rough indicators of discrimination power of the three 653 feature types independent of a specific classifier. As 654 depicted in the figure, the MSFs consistently exhibit con-655 siderably better discrimination power than the other two 656 spectral feature types, whose FDR curves are relatively 657 close. Speaker normalization is shown to be beneficial, as 658 it boosts all the FDR curves. However, selecting features 659 based solely on FDR can be hazardous, as it only evaluates 660 features individually. Effects such as feature correlation 661 and classifier properties have not been taken into account. 662 Therefore, a subsequent step to find better feature combi-663 664 nations is necessary, as performed by the second stage of our feature selection scheme. 665

The numeric results of applying SFS and LDA tech-666 niques to the FDR screened feature pools are detailed in 667 Table 2 with SVMs employed for classification and the 668 results averaged over the 10 cross-validation trials. The 669 average recognition rate is measured as the number of sam-670 ples from all emotions correctly recognized divided by the 671 672 total number of samples. For SFS, the results shown in Table 2 are for the number of features that yields the best 673 average performance. This best-performing number of fea-674 675 tures selected by SFS and the feature pool size (after FDR screening) are given by "X" and "Y", respectively, in the 676 format of "X/Y" as shown in the table. Because the pre-677 screened PLP feature pool consists of 51 features only, 678 SFS is terminated at 50 features for all feature types so 679 680 that, for fair comparison, the maximum number of features that can be selected from each feature type is the same. For 681 LDA, all the six transformed features are used as men-682 tioned in Section 5.1.1. 683

As shown in Table 2, the proposed MSFs achieve the 684 best accuracy in most emotions and in overall perfor-685 mance, reaching up to 85.6% average recognition rate 686 (using LDA with SN). Applying SN improves both SFS 687 and LDA performance for all features. It is also interesting 688 to see that in these tests, using six LDA transformed fea-689 tures delivers even higher accuracy than using dozens of 690 SFS features, indicating that the effective reduction of fea-691 ture dimensionality offered by LDA indeed contributes to 692 recognition performance. The average recognition rate 693 for the three feature types are further depicted in Fig. 10, 694 as a function of the number of features selected by SFS. 695 It is clear from Fig. 10 that the MSFs consistently outper-696 form MFCC and PLP features, irrespective of SN, though 697 performing SN boosts the accuracy curves more notably 698 for short-term spectral features than for the MSFs. More-699 over, as indicated by the figure, MFCC features appear to 700 be more suitable for emotion recognition relative to PLP 701 features, despite FDR results suggesting the two feature 702 types to be comparable. Such finding resonates with the 703 aforementioned fact that additional procedures are needed 704 to explore feature combinations after FDR pre-screening. 705



Fig. 10. Recognition results for MSF, MFCC, and PLP features selected by SFS (Berlin).

Please cite this article in press as: Wu, S. et al., Automatic speech emotion recognition using modulation spectral features, Speech Comm. (2010), doi:10.1016/j.specom.2010.08.013

SPECOM 1927

ARTICLE IN PRESS

10

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

706 5.1.3. Comparison with prosodic features

7 September 2010 Disk Used

Due to the widespread use of prosodic features, it is 707 important to study the contribution of the proposed MSFs 708 as complementary features. Comparisons are hence per-709 formed using: (1) prosodic features only and (2) combined 710 prosodic and proposed features. Although the prosodic 711 712 features extracted in our study do not exhaust all possibilities, they do cover important aspects of prosodic informa-713 tion. Consequently, recognition results for this typical 714 prosodic feature set can serve as a rule of thumb for the 715 underlying extensive prosodic feature space. 716

The average FDR curves of the two feature types and 717 their combinations are shown in Fig. 11, where the label 718 "PROS" stands for prosodic features. As can be seen from 719 the figure, the proposed features outperform prosodic fea-720 tures in terms of FDR scores, and the discrimination power 721 of the prosodic feature pool can be substantially enhanced 722 after the inclusion of MSFs. Both feature types benefit con-723 siderably from SN. The recognition rate trajectories using 724



Fig. 11. Average FDR curves for prosodic features, MSFs, and their combination (Berlin).



Fig. 12. Comparison between prosodic features and their combination with MSFs (Berlin).

Table 3

Top 10 features for prosodic,	proposed,	and	combined	features	as ranked
by AFR (Berlin).					

Rank	Feature	AFR	Rank	Feature	AFR
Prosod	lic features				
1	TEO	2.4	6	$Q_3 - Q_2$ of pitch	8.4
2	Mean syllable duration	3.3	7	Kurtosis of intensity	8.6
2	$Q_3 - Q_2$ of delta- pitch	3.3	8	Q_3 of delta-pitch	9.1
4	Slope of pitch	7.3	9	Minimum of intensity	9.8
5	Q_1 of delta-pitch	8.3	9	ZCR	9.8
Propos	sed features				
1	Mean of $\Phi_{1,k}(3)$	2.3	6	Mean of $\Phi_{6,k}(2)$	7.0
2	Mean of $C_k(4,5)$	5.0	7	Mean of $\Phi_{1,k}(2)$	7.1
3	Mean of $\Phi_{3,k}(1)$	5.6	8	Mean of $\Phi_{4,k}(4)$	7.7
4	Mean of $\Phi_{3,k}(3)$	6.0	9	Mean of $\Phi_{3,k}(5)$	8.7
5	Mean of $\Phi_{5,k}(2)$	6.4	10	Mean of $C_k(1,5)$	9.1
Combi	ned features				
1	Mean of $\Phi_{1,k}(3)$	3.0	6	Mean of $\Phi_{6,k}(2)$	7.8
2	Mean syllable duration	4.5	7	Q_1 of pitch	9.0
3	Mean of $\Phi_{3,k}(3)$	6.0	8	Q_1 of delta-pitch	9.1
4	Mean of $\Phi_{3k}(1)$	6.1	9	Slope of pitch	9.2
5	Mean of $\Phi_{5,k}(2)$	7.2	10	Mean of $C_k(4,5)$	9.4

features selected by SFS are illustrated in Fig. 12, where the contribution of the MSFs is evident.

The top features (no SN) selected by SFS from prosodic features, MSFs, and their combination are presented in Table 3, by means of average feature rank (AFR). Denote the candidate feature pool as *F*. The AFR of the *u*th feature $f_u \in F$ given *R* cross-validation trials is calculated as:

$$AFR(f_u) = \frac{1}{R} \sum_{r=1}^{R} \text{ rank of } f_u \text{ in the } r \text{th trial.}$$
(14)

If f_u is not selected in a trial, its rank is replaced by a penalty value *P*. The AFR tables here are produced by 736



Fig. 13. Comparison between different combinations of spectral and prosodic features (Berlin).

Please cite this article in press as: Wu, S. et al., Automatic speech emotion recognition using modulation spectral features, Speech Comm. (2010), doi:10.1016/j.specom.2010.08.013

726 727 728

729

730

731

732

734

ARTICLE IN PRESS

11

7 September 2010 Disk Used

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

selecting the top 10 features in each trial with the penalty 737 738 value P set to 11. Features with small AFR values, such as TEO and $\Phi_{1,\ell}(3)$, are consistently top ranked across 739 the cross-validation trials. In practice, such results can be 740 741 used to pick features when forming a final feature set to train the classifier. We also compiled AFR tables for fea-742 743 tures with SN (not shown). It is observed that nearly half of the features in Table 3 appear in the SN case as well, 744 though with different ranks, and several features remain 745 top ranked regardless of SN, such as TEO, mean syllable 746 duration, $\Phi_{1,k}(3)$ and $\Phi_{3,k}(1)$. 747

The short-term spectral features are also considered for comparison. SFS results with different combinations of spectral feature type and prosodic features are given in



Fig. 14. Average recognition performance of N_{best} for different combinations of spectral and prosodic features (Berlin).

Fig. 13, where the proposed features achieve the highest 751 recognition rate, either with SN (MSF: 87.7%, MFCC: 752 86.5%, PLP: 87.3%) or without SN (MSF: 85.4%, MFCC: 753 84.1%, PLP: 81.9%). Although the best performance of 754 different feature combinations might seem close, the advan-755 tage of the proposed features is most notable in Fig. 14. 756 The label N_{best} denotes that the N_{best} ($1 \le N_{best} \le 50$) high-757 est recognition rates (among the 50 recognition rates 758 obtained by selecting 1-50 features using SFS) are chosen, 759 with their average being calculated and depicted in the fig-760 ure. As shown in Fig. 14, the MSFs always furnish the 761 highest average recognition rate relative to MFCC and 762 PLP features, irrespective of SN. But among the three spec-763 tral feature types evaluated by SFS, PLP features (com-764 bined with prosodic features) turn out to receive the 765 largest performance gain from SN. 766 767

Table 4 details the results achieved by the combined features, with recognition rate shown for each emotion. LDA results are included as well. The row labeled " \downarrow %" indicates the percentage reduction of error rate obtained by adding the MSFs to prosodic features, which is calculated as:

772 773

768

769

770

771

$$\% = \frac{\mathbf{R}\mathbf{R}_{\text{PROS}+\text{MSF}} - \mathbf{R}\mathbf{R}_{\text{PROS}}}{1 - \mathbf{R}\mathbf{R}_{\text{PROS}}} \times 100\%, \tag{15}$$

where "RR" represents recognition rate. Similar to the case 776 where spectral features are evaluated individually, the 777 MSFs achieve the highest overall accuracy when combined 778 with prosodic features, and up to 91.6% recognition rate 779 can be obtained using LDA with SN. Applying LDA with 780 SN also gives the best recognition performance for 781

Table 4

Recognition results for prosodic and combined features (Berlin); boldface indicates the best performance in each test.

Test	Feature	Method	SN	Recogni	tion rate (%)						Average
				Anger	Boredom	Disgust	Fear	Joy	Neutral	Sadness	
#1	PROS (50/65)	SFS	No	89.8	87.7	73.9	84.1	59.2	79.8	83.9	81.1
	PROS + MSF (48/139)			94.5	88.9	76.1	84.1	57.8	89.9	96.8	85.4
	PROS + MFCC (38/157)			93.7	88.9	71.7	85.5	56.3	89.9	90.3	84.1
	PROS + PLP (45/116)			90.6	84.0	71.7	78.3	59.2	89.9	88.7	81.9
	↓ %			46.1	9.8	8.4	0.0	-3.4	50.0	80.1	22.8
#2	PROS	LDA	No	87.4	82.7	78.3	79.7	49.3	82.3	83.9	78.7
	PROS + MSF			90.6	92.6	87.0	82.6	62.0	87.3	88.7	85.0
	PROS + MFCC			84.3	87.7	93.5	79.7	62.0	88.6	91.9	83.6
	PROS + PLP			87.4	90.1	89.1	68.1	57.8	84.8	88.7	81.3
	↓%			25.4	57.2	40.1	14.3	25.1	28.3	29.8	29.6
#3	PROS (41/64)	SFS	Yes	92.9	91.4	78.3	81.2	56.3	87.3	90.3	83.9
	PROS + MSF (48/137)			94.5	87.7	82.6	84.1	63.4	96.2	98.4	87.7
	PROS + MFCC (44/160)			93.7	92.6	82.6	84.1	66.2	84.8	95.2	86.5
	PROS + PLP (44/115)			96.1	92.6	78.3	82.6	63.4	92.4	95.2	87.3
	↓ %			22.5	-43.0	19.8	15.4	16.3	70.1	83.5	23.6
#4	PROS	LDA	Yes	89.8	85.2	80.4	87.0	62.0	93.7	93.6	85.2
	PROS + MSF			93.7	96.3	91.3	89.9	73.2	94.9	100	91.6
	PROS + MFCC			88.2	87.7	89.1	84.1	69.0	89.9	91.9	85.8
	PROS + PLP			85.8	87.7	78.3	84.1	67.6	93.7	91.9	84.7
	↓ %			38.2	75.0	55.6	22.3	29.5	19.0	100	43.2

ARTICLE IN PRESS

7 September 2010 Disk Used

12

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

Table	5
-------	---

Confusion matrix for	using only	prosodic features	(Berlin).
----------------------	------------	-------------------	-----------

Emotion	Anger	Boredom	Disgust	Fear	Joy	Neutral	Sadness	Rate (%)
Anger	114	0	1	3	8	1	0	89.8
Boredom	0	69	1	0	0	8	3	85.2
Disgust	0	1	37	2	0	5	1	80.4
Fear	4	0	0	60	3	2	0	87.0
Joy	19	0	1	2	44	5	0	62.0
Neutral	1	2	1	0	0	74	1	93.7
Sadness	0	3	0	0	0	1	58	93.6
Precision (%)	82.6	92.0	90.2	89.6	80.0	77.1	92.1	

Table 6 Confusion matrix for using prosodic and proposed features (Berlin).

Emotion	Anger	Boredom	Disgust	Fear	Joy	Neutral	Sadness	Rate (%)
Anger	119	0	1	1	6	0	0	93.7
Boredom	0	78	0	0	0	3	0	96.3
Disgust	0	0	42	1	1	2	0	91.3
Fear	2	0	0	62	3	2	0	89.9
Joy	12	0	1	3	52	3	0	73.2
Neutral	0	3	1	0	0	75	0	94.9
Sadness	0	0	0	0	0	0	62	100
Precision (%)	89.5	96.3	93.3	92.5	83.9	88.2	100	

prosodic features (85.2%). For MFCC and PLP features,
however, superior results are observed for SFS with SN.

Two confusion matrices are shown in Tables 5 and 6 784 (left-most column being the true emotions), for the best 785 recognition performance achieved by prosodic features 786 alone and combined prosodic and proposed features 787 (LDA + SN), respectively. The *rate* column lists per class 788 recognition rates, and *precision* for a class is the number 789 of samples correctly classified divided by the total number 790 of samples classified to the class. We can see from the con-791 fusion matrices that adding MSFs contributes to improv-792 ing the recognition and precision rates of all emotion 793 categories. It is also shown that most emotions can be cor-794 rectly recognized with above 89% accuracy, with the excep-795 796 tion of *joy*, which forms the most notable confusion pair with anger, though they are of opposite valence in the acti-797 vation-valence emotion space (Cowie et al., 2001). This 798 might be due to the fact that activation is more easily rec-799 ognized by machine than valence, as indicated by the 800 regression results for the emotion primitives on the VAM 801 database presented in Section 5.2. 802

As aforementioned, the cross-validation scheme used so 803 far is not entirely speaker-independent. We further investi-804 gate the effect of speaker dependency for SER by doing 805 806 "leave-one-speaker-out" (LOSO) cross-validation, wherein the training set does not contain a single instance of the 807 speaker in the test set. LOSO results with different features 808 are presented in Table 7 and compared with the previous 809 speaker-dependent 10-fold results in Tables 2 and 4 (simply 810 denoted as "10-fold" in the table). As shown in the table, 811 emotion recognition accuracy under the more stringent 812 LOSO condition is lower than when test speakers are rep-813

resented in the training set. This expected behavior applies 814 to all feature types. When tested alone, the MSFs clearly 815 outperform MFCC and PLP features. When combined 816 with prosodic features, MFCC features yield the best per-817 formance if SN is applied; otherwise, the MSFs still prevail. 818 As SN is not applied in typical real-life applications, the 819 proposed features might be more suitable for these more 820 realistic scenarios. 821

However, it should also be noted that the Berlin database has limited phonetic content (10 acted sentences), 823 hence limiting the generalizability of the obtained results. 824

Ta	ble	7
1 11	010	'

Recognition results for LOSO cross-validation	on (Berlin)
---	-------------

Feature	Recognition rate (%)							
	SFS		LDA	LDA				
	10-fold	LOSO	10-fold	LOSO				
MSF	79.6	74.0	81.3	71.8				
MFCC	76.5	65.6	77.9	66.0				
PLP	71.2	63.2	71.4	65.0				
MSF (SN)	82.8	78.1	85.6	79.1				
MFCC (SN)	78.5	71.8	78.7	72.3				
PLP (SN)	75.1	72.3	77.8	72.5				
PROS	81.1	75.0	78.7	71.2				
PROS + MSF	85.4	78.1	85.0	76.3				
PROS + MFCC	84.1	75.1	83.6	75.5				
PROS + PLP	81.9	75.3	81.3	72.1				
PROS (SN)	83.9	83.0	85.2	80.2				
PROS + MSF(SN)	87.7	83.2	91.6	80.9				
PROS + MFCC (SN)	86.5	85.8	85.8	82.4				
PROS + PLP (SN)	87.3	84.3	84.7	78.7				

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

It is also useful to briefly review performance figures 825 reported on the Berlin database by other works. Although 826 the numbers cannot be directly compared due to factors 827 such as different data partitioning, they are still useful for 828 829 general benchmarking. Unless otherwise specified, the results cited here are achieved for recognizing all seven 830 831 emotions with UL features. For works that do not use 832 speaker normalization, 86.7% recognition rate is achieved under 10-fold cross-validation in (Schuller et al., 2006), 833 by using around 4000 features. The accuracy is slightly 834 improved to 86.9% after optimizing the feature space, but 835 the dimensionality of the optimized space is not reported. 836 In (Lugger and Yang, 2008), 88.8% accuracy is achieved 837 by employing a three-stage classification scheme, but based 838 on recognition of six emotions only (no disgust). Among 839 the works that use speaker normalization, 83.2% recogni-840 tion rate is obtained in a leave-one-speaker-out experiment 841 (Vlasenko et al., 2007) by extracting around 1400 acoustic 842 843 features for data mining. However, no information is provided about the final number of features used. The accu-844 racy is further improved to 89.9% by integrating both 845 846 UL and FL features.

847 5.2. Experiment II: continuous emotion regression

The well-established descriptive framework that uses 848 discrete emotions offers intuitive emotion descriptions 849 and is commonly used. The combination of basic emotion 850 categories can also serve as a convenient representation of 851 the universal emotion space (Ekman, 1999). However, 852 recent research efforts also show an increasing interest in 853 dimensional representations of emotions for SER (Grimm 854 et al., 2007a; Grimm et al., 2007b; Wollmer et al., 2008; 855 Giannakopoulos et al., 2009). The term "dimensional" here 856 refers to a set of primary emotion attributes that can be 857 treated as the bases of a multi-dimensional emotion space. 858 wherein categorical descriptors can be situated by coordi-859 860 nates. A dimensional framework allows for gradual change within the same emotion as well as transition between emo-861 tional states. In this experiment, we recognize the three 862 continuous emotion primitives - valence, activation and 863 dominance - in the VAM database. 864

5.2.1. Continuous emotion recognition

Leave-one-out (LOO) cross-validation is used to enable comparisons with the results reported in (Grimm et al., 2007a). In this LOO test, the speaker in the test instance is also represented in the training set. Akin to the comparison framework in Section 5.1, regression is performed using (1) spectral features, (2) prosodic features, and (3) combined prosodic and spectral features (without SN). The SFS algorithm is employed to select the best features for the support vector regressors (SVRs). Experiments are carried out on three datasets: VAM I. VAM II. and VAM I + II. Ideally, LOO cross-validation on N-sample data requires SFS to be applied to all the N different training sets, each containing N-1 samples. However, since N is reasonably large here (478, 469, and 947 for VAM I, II, and I + II, respectively), including the test sample hardly impacts the SFS selections. Thus in each experiment, feature selection is carried out using all the samples of the corresponding dataset. LOO is then used to re-train the regressor using the N-1 samples in each validation trial and test the remaining sample.

Correlation coefficient r and mean absolute error e are the two performance measures to be calculated. For two sequences x_n and y_n of the same length, the correlation coefficient is calculated using Pearson's formula:

$$r = \frac{\sum_{n} (x_{n} - \bar{x})(y_{n} - \bar{y})}{\sqrt{\sum_{n} (x_{n} - \bar{x})^{2} \sum_{n} (y_{n} - \bar{y})^{2}}},$$
(16)

where $\bar{x}(\bar{y})$ is the average of $x_n(y_n)$. Mean absolute error is used, identical to the error measure employed in (Grimm et al., 2007a).

Mean correlations, averaged over the three primitives, are shown in Fig. 15 for the spectral features. Unlike the emotion classification results for the Berlin database, MFCC features furnish better regression performance on the VAM datasets as suggested by the figure. A closer examination (see also Table 8) reveals that the gain of MFCC features over MSFs is mainly due to the former's better performance on the *valence* primitive, especially for the VAM II dataset. Such results suggest that MFCC features are more competent than the proposed features at





Please cite this article in press as: Wu, S. et al., Automatic speech emotion recognition using modulation spectral features, Speech Comm. (2010), doi:10.1016/j.specom.2010.08.013

13

865

866

867

868

869

870

871

872

873

874

875

876

877

878

879

880

881

882

883

884

885

886

887

888

889 890

892

893

894

895

896

897

898

899

900

901

902

903

904

Table 0

7 September 2010 Disk Used

ARTICLE IN PRESS

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

Regression results for	continuous emotion	s on the VAM	database using	MSF, MFCC	, and PLP features.

Dataset	Feature	Correlation (r)			Absolute error (e)			Average	
		Valence	Activation	Dominance	Valence	Activation	Dominance	\overline{r}	ē
VAM I	MSF	0.60	0.86	0.81	0.11	0.15	0.15	0.76	0.14
	MFCC	0.65	0.87	0.81	0.11	0.14	0.15	0.78	0.13
	PLP	0.61	0.86	0.79	0.11	0.16	0.16	0.75	0.14
VAM II	MSF	0.32	0.74	0.66	0.15	0.16	0.16	0.57	0.16
	MFCC	0.46	0.74	0.68	0.14	0.16	0.15	0.63	0.15
	PLP	0.26	0.63	0.59	0.15	0.18	0.16	0.49	0.16
VAM I + II	MSF	0.46	0.80	0.75	0.13	0.17	0.16	0.67	0.15
	MFCC	0.52	0.79	0.74	0.13	0.17	0.16	0.68	0.15
	PLP	0.42	0.75	0.70	0.13	0.18	0.17	0.62	0.16

determining the positive or negative significance of emo-906 tions (i.e. valence). On the other hand, the two feature types 907 provide very close performance for recognizing activation 908 and *dominance* as indicated by the correlation curves for 909 individual primitives (not shown). PLP features give infe-910 rior regression results on VAM II and VAM I+II, again 911 becoming the spectral feature type that yields the worst 912 SER outcomes. The highest mean correlation achieved by 913 each feature type in Fig. 15 is further interpreted in Table 914 8 by showing the regression performance for individual 915 primitives. Comparing the three primitives, activation 916 receives the best correlations (0.63-0.87), while valence 917 shows significantly lower correlations (0.26-0.65) for all 918 features. It is also observed that the absolute regression 919 errors between the different spectral feature types are quite 920 close. 921

Regression of prosodic and combined features is consid-922 ered in Fig. 16 and Table 9. The machine recognition and 923 human subjective evaluation results given in (Grimm 924 et al., 2007a) are also included in Table 9 for reference. 925 Note that in (Grimm et al., 2007a), only the standard devi-926 ation of subjective scores is presented for each primitive. 927 The error is the standard deviation minus a constant bias 928 term that depends on the number of evaluators, which 929 can be inferred from the paper. As shown in the figure 930 and table, adding spectral features to prosodic features 931 932 improves the correlations for all primitives on all datasets, and slightly reduces the estimation errors in some cases. 933

The MFCC features (combined with prosodic features) still 934 deliver the best performance, followed by the proposed fea-935 tures. Again the advantage of the MFCC features over 936 MSFs is on recognizing valence. The performance gaps 937 between MSFs and MFCC features in Table 8 appear nar-938 rowed in Table 9, indicating that the prosodic features 939 enhance MSF performance more than MFCC perfor-940 mance, mainly because the prosodic features contribute 941 more to improving *valence* recognition for the MSF case, 942 even though overall MFCC performance is slightly better. 943

The recognition tendency for the primitives in Table 9 is 944 the same as the trend observed in Table 8. The primitive 945 activation is best estimated with up to 0.90 correlation 946 achieved on VAM I, followed by dominance whose regres-947 sion performance is more moderate. Even though the inclu-948 sion of spectral features improves the correlations for 949 valence, the attained values are relatively low for all feature 950 combinations. Nevertheless, even human evaluations give 951 poor correlations for recognizing valence compared to the 952 other two primitives. 953

Overall, the recognition system using combined features 954 yields higher correlations and smaller estimation errors 955 compared to the machine recognition results in (Grimm 956 et al., 2007a). The performance of the proposed recognition 957 system appears to be somewhat superior to human assess-958 ment; this however merely reflects the capability of 959 machines to be more consistent (but not more accurate) 960 than humans in performing the emotion labeling task. In 961





Table 9

7 September 2010 Disk Used

ARTICLE IN PRESS

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

Regression results for continuous emotions on the VAM database using prosodic and combined features.

Dataset	Feature	Correlation (<i>r</i>)		Absolute error (e)			Average		
		Valence	Activation	Dominance	Valence	Activation	Dominance	\bar{r}	ē
VAM I	PROS only	0.58	0.85	0.82	0.11	0.16	0.15	0.75	0.14
	PROS + MSF	0.66	0.90	0.86	0.10	0.13	0.13	0.81	0.12
	PROS + MFCC	0.66	0.90	0.86	0.10	0.13	0.13	0.81	0.12
	PROS + PLP	0.62	0.90	0.85	0.11	0.13	0.14	0.79	0.13
	Grimm et al. (2007a)	N/A	N/A	N/A	N/A	N/A	N/A	0.71	0.27
	Human	0.49	0.78	0.68	0.17	0.25	0.20	0.65	0.21
VAM II	PROS only	0.38	0.70	0.64	0.14	0.17	0.15	0.57	0.15
	PROS + MSF	0.48	0.78	0.73	0.14	0.15	0.14	0.66	0.14
	PROS + MFCC	0.54	0.79	0.73	0.13	0.15	0.14	0.69	0.14
	PROS + PLP	0.45	0.75	0.70	0.14	0.16	0.14	0.63	0.15
	Grimm et al. (2007a)	N/A	N/A	N/A	N/A	N/A	N/A	0.43	0.23
	Human	0.48	0.66	0.54	0.11	0.19	0.14	0.56	0.15
VAM I+II	PROS only	0.47	0.77	0.75	0.13	0.17	0.15	0.66	0.15
	PROS + MSF	0.55	0.82	0.80	0.13	0.15	0.14	0.72	0.14
	PROS + MFCC	0.56	0.83	0.80	0.12	0.15	0.14	0.73	0.14
	PROS + PLP	0.52	0.82	0.78	0.13	0.16	0.15	0.71	0.15
	Grimm et al. (2007a)	N/A	N/A	N/A	N/A	N/A	N/A	0.60	0.24
	Human	0.49	0.72	0.61	0.14	0.22	0.17	0.61	0.18

962 (Grimm et al., 2007b), good estimation results are also achieved for activation and dominance on VAM I+II, 963 but valence is still poorly estimated with 0.46 correlation 964 reported. This corroborates the poor anger vs. joy classifi-965 cation results in Tables 5 and 6, as the two emotions are 966 with the same activation level but opposite valence. Since 967 it has also been shown that anger vs. sadness, emotions 968 969 with opposite activation but similar valence, can be sepa-970 rated very well, both the Berlin (discrete and acted) and VAM (continuous and natural) databases show the ten-971 972 dency of the SER systems to recognize activation more 973 accurately than valence.

974 Moreover, as might also be noticed from Figs. 15 and 16 as well as Tables 8 and 9, regression performance on VAM 975 976 I is always superior to the performance on VAM II, regardless of the feature types. This is because VAM I contains 977 978 utterances from "very good" speakers who are character-979 ized by a high level of activity and a wide variety of emotions, while VAM II consists of utterances from "good" 980 speakers that, though possessing high activity as well, pro-981 duce a smaller scope of emotions (e.g. anger only) (Grimm 982 983 et al., 2008). VAM I + II, as a combination of VAM I and VAM II, exhibits recognition performance intermediate 984 between VAM I and VAM II. 985

986 5.2.2. Cross-database evaluation

In (Grimm et al., 2007a), the authors mapped the con-987 tinuous-valued estimates of the emotion primitives into dis-988 crete emotion categories on the EMA corpus (Lee et al., 989 990 2005). In this section, a similar experiment is performed 991 to apply the continuous primitives to classify the discrete 992 emotions in the Berlin database. More specifically, since the regression performance varies for the three primitives 993 994 as shown in the previous experiment, several binary classification tasks are employed here to further evaluate the primitives separately.

First, three SVRs are trained on the VAM I + II dataset using the combined prosodic and proposed features, one for each of the three primitives. Each SVR uses the set of features selected by SFS on VAM I + II that yield the highest correlation. The Berlin database is then used as a separate source for evaluation, where each file is assigned three predicted primitive values by the three trained regressors. The three estimated primitive values are then used as features for classifying the seven emotions, yielding an overall recognition rate of 52.7% (49.9%) under 10-fold cross-validation with (without) SN. Even though this accuracy is unattractive, it is still better than the 14% random chance, indicating that the continuous primitives do convey useful information about the discrete emotion classes in the Berlin database. The low recognition rate could be due to inadequacy of using the three primitives. Also, no "mutual information" relating the emotion data of the two databases is available. Such information could have been produced by having the same human subjects rate both databases on both the discrete and primitive scales.

Three two-class classification tasks are designed to test the primitive features: (1) *anger* vs. *joy*, (2) *anger* vs. *sadness*, and (3) *anger* vs. *fear*, which mainly involve (though not limited to) the recognition of *valence*, *activation*, and *dominance*, respectively. Ideally, if the three primitives were well recognized, *valence*, *activation*, and *dominance* would be the features providing the best recognition performance for tasks (1), (2), and (3), respectively.

Recognition results are listed in Table 10 where SN has been applied (in the same way as did in Section 5.1). As can be seen from the table, high classification accuracy is achieved when discriminating *anger* vs. *sadness* (100%) and *anger* vs. *fear* (83.7%) using only one primitive feature,

1029

16

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

Table 10

Binary classification results with three primitive features (cross-database).

	Valence (%)	Activation (%)	Dominance (%)	All (%)
Anger vs. joy	63.6	61.6	61.1	58.6
Anger vs. sadness	65.6	100	99.5	99.0
Anger vs. fear	64.8	79.6	83.7	81.6

namely activation and dominance, respectively. Although 1030 the unrealistic 100% accuracy may be partly due to the 1031 acted emotion in the Berlin database, the result still corrob-1032 orates previous results that showed the activation primitive 1033 to be well recognized by machine. Combining all three 1034 primitive features, however, does not improve the recogni-1035 tion performance. One notable difficulty, as expected, 1036 arises when using valence to classify anger and joy, which 1037 1038 are with similar activations and opposite valence, resulting in 63.6% recognition rate only. These cross-database 1039 results reinforce our previous findings that activation and 1040 valence are primitives with the best and worst estimation 1041 performance, respectively. In addition, the seven-class 1042 FDR scores for the three primitive features are valence: 1043 1044 0.6 (0.4), activation: 4.4 (3.7), and dominance: 3.9 (3.3) with (without) SN, which again indicate that the valence feature 1045 has the lowest discrimination power. 1046

6. Conclusion 1047

This work presents novel MSFs for the recognition of 1048 human emotions in speech. An auditory-inspired ST repre-1049 sentation is acquired by deploying an auditory filterbank as 1050 well as a modulation filterbank, to perform spectral decom-1051 position in the conventional acoustic frequency domain 1052 and in the modulation frequency domain, respectively. 1053 The proposed features are then extracted from this ST rep-1054 resentation by means of spectral measures and linear pre-1055 1056 diction parameters.

The MSFs are evaluated first on the Berlin database to 1057 classify seven discrete emotions. Typical short-term spec-1058 tral features and prosodic features are extracted to bench-1059 mark the proposed features. Simulation results show that 1060 the MSFs outperform MFCC and PLP features when each 1061 feature type is used solely, in terms of both FDR scores and 1062 recognition accuracy. The proposed features are also 1063 1064 shown to serve as powerful additions to prosodic features, as substantial improvement in recognition accuracy is 1065 achieved once prosodic features are combined with the 1066 MSFs, with up to 91.6% overall recognition accuracy 1067 attained. In a LOSO cross-validation test, the MSFs give 1068 1069 superior performance except in the case when speaker normalization is applied; MFCC combined with prosodic fea-1070 1071 tures outperform MSFs combined with prosodic features.

1072 Besides the classic discrete emotion classification, continuous emotion estimation is further investigated in this 1073 study. MFCC features are shown to deliver the best results 1074 in the regression experiments. However, the apparent gain 1075

of MFCC features over MSFs is mainly due to the former's 1076 higher correlation with valence. For the other two primi-1077 tives, the performance of the proposed and MFCC features 1078 are comparable. PLP features are found to deliver gener-1079 ally inferior outcomes relative to the other two spectral fea-1080 ture types. Combining spectral and prosodic features 1081 further improves the regression results. Promising perfor-1082 mance is achieved for estimating activation and dominance, 1083 but lower performance is observed with valence, a trend 1084 also reported in other continuous emotion recognition 1085 studies. Moreover, the continuous emotion primitives are 1086 applied to classify discrete emotions. Among the three 1087 primitive features, activation and dominance are shown to 1088 be useful for classifying discrete emotions, but further 1089 investigation is needed for the valence parameter to achieve 1090 practical performance figures. 1091

Combining the performance results for the Berlin and 1092 VAM databases, the new MSFs appear to offer equally 1093 competitive performance with respect to prosodic and 1094 MFCC features. Over the decades, MFCCs have been well 1095 optimized and demonstrated good performance for auto-1096 matic speech recognition applications. However, features 1097 extracting modulation spectral information have recently 1098 demonstrated promising performance for various applica-1099 tions in speech processing (e.g. Falk and Chan, 2008, 1100 2010a,). This paper has demonstrated the potential and 1101 promise of the MSFs for emotion recognition. With possi-1102 ble refinement in future work, the performance of modula-1103 tion domain features could be further improved. Hence, 1104 further research on the use of MSFs for SER can be 1105 beneficial. 1106

Acknowledgements

The authors would like to thank Dr. Michael Grimm for 1108 providing the VAM database and the reviewers for helpful 1109 improvement of the paper. 1110

References

- Abelin, A., Allwood, J., 2000. Cross linguistic interpretation of emotional prosody. In: Proc. ISCA ITRW on Speech and Emotion, pp. 110-113.
- Aertsen, A., Johannesma, P., 1980. Spectro-temporal receptive fields of auditory neurons in the grass frog. I. Characterization of tonal and natural stimuli. Biol. Cybernet. 38, 223-234.
- Atlas, L., Shamma, S., 2003. Joint acoustic and modulation frequency. EURASIP J. Appl. Signal Process., 668-675.
- Barra, R., Montero, J., Macias-Guarasa, J., D'Haro, L., San-Segundo, R., Cordoba, R., 2006. Prosodic and segmental rubrics in emotion identification. In: Proc. Internat. Conf. on Acoustics, Speech and Signal Processing, Vol. 1, pp. 1085-1088.
- Batliner, A., Steidl, S., Schuller, B., Seppi, D., Laskowski, K., Vogt, T., Devillers, L., Vidrascu, L., Amir, N., Kessous, L., Aharonson, V., 2006. Combining efforts for improving automatic classification of emotional user states. In: Proc. IS-LTC, pp. 240-245.
- Bishop, C., 2006. Pattern Recognition and Machine Learning. Springer, New York.
- Burkhardt, F., Paeschke, A., Rolfes, M., Sendlmeier, W., Weiss, B., 2005. A database of German emotional speech. In: Proc. Interspeech, pp. 1517-1520.

Please cite this article in press as: Wu, S. et al., Automatic speech emotion recognition using modulation spectral features, Speech Comm. (2010), doi:10.1016/j.specom.2010.08.013

1107

1111

1128 1129

1130

ARTICLE IN PRESS

S. Wu et al. | Speech Communication xxx (2010) xxx-xxx

- Busso, C., Lee, S., Narayanan, S., 2009. Analysis of emotionally salient aspects of fundamental frequency for emotion detection. IEEE Trans.
 Audio Speech Language Process. 17, 582–596.
- Chang, C.-C., Lin, C.-J., 2009. LIBSVM: A library for support vector machines. Tech. rep., Department of Computer Science, National Taiwan University. Software available at: <<u>http://www.csie.ntu.e-</u> du.tw/cjlin/libsvm>.
- Chih, T., Ru, P., Shamma, S., 2005. Multiresolution spectrotemporal analysis of complex sounds. J. Acoust. Soc. Amer. 118, 887–906.
- 1141 Clavel, C., Vasilescu, I., Devillers, L., Richard, G., Ehrette, T., 2008. Feartype emotion recognition for future audio-based surveillance systems.
 1143 Speech Commun. 50, 487–503.
- Cowie, R., Douglas-Cowie, E., 1996. Automatic statistical analysis of the signal and prosodic signs of emotion in speech. In: Proc. Internat.
 Conf. on Spoken Language Processing, Vol. 3, pp. 1989–1992.
- Cowie, R., Douglas-Cowie, E., Tsapatsoulis, N., Votsis, G., Kollias, S.,
 Fellenz, W., Taylor, J., 2001. Emotion recognition in humancomputer interaction. IEEE Signal Process. Mag. 18, 32–80.
- Davis, S., Mermelstein, P., 1980. Comparison of parametric representations for monosyllabic word recognition in continuously spoken sentences. IEEE Trans. Audio Speech Language Process. 28, 357–366.
- Depireux, D., Simon, J., Klein, D., Shamma, S., 2001. Spectro-temporal
 response field characterization with dynamic ripples in ferret primary
 auditory cortex. J. Neurophysiol. 85, 1220–1234.
- 1156 Douglas-Cowie, E., Campbell, N., Cowie, R., Roach, P., 2003. Emotional
 1157 speech: Towards a new generation of databases. Speech Commun. 40,
 1158 33–60.
- 1159 Ekman, P., 1999. Basic Emotions. John Wiley, New York, pp. 45-60.
- Ewert, S., Dau, T., 2000. Characterizing frequency selectivity for envelope fluctuations. J. Acoust. Soc. Amer. 108, 1181–1196.
- Falk, T.H., Chan, W.-Y., 2008. A non-intrusive quality measure of dereverberated speech. In: Proc. Internat. Workshop for Acoustic Echo and Noise Control.
- Falk, T.H., Chan, W.-Y., 2010a. Modulation spectral features for robust far-field speaker identification. IEEE Trans. Audio Speech Language Process. 18, 90–100.
- Falk, T.H., Chan, W.-Y., 2010b. Temporal dynamics for blind measurement of room acoustical parameters. IEEE Trans. Instrum. Meas. 59,
 978–989.
- 1171 Fletcher, H., 1940. Auditory patterns. Rev. Modern Phys. 12, 47–65.
- Giannakopoulos, T., Pikrakis, A., Theodoridis, S., 2009. A dimensional approach to emotion recognition of speech from movies. In: Proc. Internat. Conf. on Acoustics, Speech and Signal Processing, pp. 65–68.
- Glasberg, B., Moore, B., 1990. Derivation of auditory filter shapes from notched-noise data. Hearing Res. 47, 103–138.
- Grimm, M., Kroschel, K., Mower, E., Narayanan, S., 2007a. Primitivesbased evaluation and estimation of emotions in speech. Speech Commun. 49, 787–800.
- Grimm, M., Kroschel, K., Narayanan, S., 2007b. Support vector regression for automatic recognition of spontaneous emotions in speech. In: Proc. Internat. Conf. on Acoustics, Speech and Signal Processing, Vol. 4, pp. 1085–1088.
- Grimm, M., Kroschel, K., Narayanan, S., 2008. The Vera am Mittag
 German audio-visual emotional speech database. In: Proc. Internat.
 Conf. on Multimedia & Expo, pp. 865–868.
- Gunes, H., Piccard, M., 2007. Bi-modal emotion recognition from expressive face and body gestures. J. Network Comput. Appl. 30, 1334–1345.
- Hermansky, H., 1990. Perceptual linear predictive (PLP) analysis of
 speech. J. Acoust. Soc. Amer. 87, 1738–1752.
- Hsu, C.-C., Chang, C.-C., Lin, C.-J., 2007. A practical guide to support vector classification. Tech. rep., Department of Computer Science, National Taiwan University.
- Intl. Telecom. Union, 1993. Objective measurement of active speech level.
 ITU-T P.56, Switzerland.
- Intl. Telecom. Union, 1996. A silence compression scheme for G.729
 optimized for terminals conforming to ITU-T recommendation V.70.
 ITU-T G.729 Annex B, Switzerland.

- Ishi, C., Ishiguro, H., Hagita, N., 2010. Analysis of the roles and the dynamics of breathy and whispery voice qualities in dialogue speech. EURASIP J. Audio Speech Music Process., article ID 528193, 12 pages.
- Kaiser, J., 1990. On a simple algorithm to calculate the 'energy' of a signal.In: Proc. Internat. Conf. on Acoustics, Speech and Signal Processing,Vol. 1, pp. 381–384.
- Kanederaa, N., Araib, T., Hermanskyc, H., Pavel, M., 1999. On the relative importance of various components of the modulation spectrum for automatic speech recognition. Speech Commun. 28, 43–55.
- Kittler, J., 1978. Feature set search algorithms. Pattern Recognition Signal Process., 41–60.
- Lee, S., Yildirim, S., Kazemzadeh, A., Narayanan, S., 2005. An articulatory study of emotional speech production. In: Proc. Interspeech, pp. 497–500.
- Lugger, M., Yang, B., 2008. Cascaded emotion classification via psychological emotion dimensions using a large set of voice quality parameters. In: Proc. Internat. Conf. on Acoustics, Speech and Signal Processing, Vol. 4, pp. 4945–4948.
- Morgan, N., Zhu, Q., Stolcke, A., Sonmez, K., Sivadas, S., Shinozaki, T., Ostendorf, M., Jain, P., Hermansky, H., Ellis, D., Doddington, G., Chen, B., Cetin, O., Bourlard, H., Athineos, M., 2005. Pushing the envelope – aside. IEEE Signal Process. Mag. 22, 81–88.
- Mozziconacci, S., 2002. Prosody and emotions. In: Proc. Speech Prosody, pp. 1–9.
- Nwe, T., Foo, S., De Silva, L., 2003. Speech emotion recognition using hidden markov models. Speech Commun. 41, 603–623.
- Picard, R., 1997. Affective Computing. The MIT Press.
- Rabiner, L., Juang, B., 1993. Fundamentals of Speech Recognition. Prentice-Hall, Englewood Cliffs, NJ.
- Scherer, K., 2003. Vocal communication of emotion: A review of research paradigms. Speech Commun. 40, 227–256.
- Scherer, S., Schwenker, F., Palm, G., 2007. Classifier fusion for emotion recognition from speech. In: Proc. IET Internat. Conf. on Intelligent Environments, pp. 152–155.
- Schuller, B., Seppi, D., Batliner, A., Maier, A., Steidl, S., 2006. Emotion recognition in the noise applying large acoustic feature sets. In: Proc. Speech Prosody.
- Schuller, B., Batliner, A., Seppi, D., Steidl, S., Vogt, T., Wagner, J., Devillers, L., Vidrascu, L., Amir, N., Kessous, L., Aharonson, V., 2007a. The relevance of feature type for the automatic classification of emotional user states: Low level descriptors and functionals. In: Proc. Interspeech, pp. 2253–2256.
- Schuller, B., Seppi, D., Batliner, A., Maier, A., Steidl, S., 2007b. Towards more reality in the recognition of emotional speech. In: Proc. Internat. Conf. on Acoustics, Speech and Signal Processing, Vol. 4. pp. 941–944.
- Shami, M., Verhelst, W., 2007. An evaluation of the robustness of existing supervised machine learning approaches to the classification of emotions in speech. Speech Commun. 49, 201–212.
- Shamma, S., 2001. On the role of space and time in auditory processing. Trends Cogn. Sci. 5, 340–348.
- Shamma, S., 2003. Encoding sound timbre in the auditory system. IETE J. Res. 49, 193–205.
- Slaney, M., 1993. An efficient implementation of the Patterson–Holdsworth auditory filterbank. Tech. rep., Apple Computer, Perception Group.
- Sun, R., E., M., Torres, J., 2009. Investigating glottal parameters for differentiating emotional categories with similar prosodics. In: Proc. Internat. Conf. on Acoustics, Speech and Signal Processing, pp. 4509– 4512.
- Talkin, D., 1995. A robust algorithm for pitch tracking (RAPT). Elsevier Science, Speech Coding and Synthesis (Chapter 14).
- Vapnik, V., 1995. The Nature of Statistical Learning Theory. Springer, New York.
- Ververidis, D., Kotropoulos, C., 2006. Emotional speech recognition: Resources, features, and methods. Speech Commun. 48, 1162–1181.

Please cite this article in press as: Wu, S. et al., Automatic speech emotion recognition using modulation spectral features, Speech Comm. (2010), doi:10.1016/j.specom.2010.08.013

17

1200

1201

1202 1203

1204

1205

1206

1207 1208

1209

1210

1211

1212

1213

1214

1215

1216

1217

1218

1219

1220

1221

1222

1223

1224

1225

1226

1227

1228

1229

1230

1231

1232

1233

1234

1235

1236

1237

7 September 2010 Disk Used

S. Wu et al. / Speech Communication xxx (2010) xxx-xxx

- Vlasenko, B., Schuller, B., Wendemuth, A., Rigoll, G., 2007. Combining
 frame and turn-level information for robust recognition of emotions
 within speech. In: Proc. Interspeech, pp. 2225–2228.
- Wollmer, M., Eyben, F., Reiter, S., Schuller, B., Cox, C., Douglas-Cowie,
 E., Cowie, R., 2008. Abandoning emotion classes Towards contin-
- 1273 uous emotion recognition with modelling of long-range dependencies.1274 In: Proc. Interspeech, pp. 597–600.
- Wu, S., Falk, T., Chan, W.-Y., 2009. Automatic recognition of speech emotion using long-term spectro-temporal features. In: Proc. Internat. Conf. on Digital Signal Processing, pp. 1–6.
- Zhou, G., Hansen, J., Kaiser, J., 2001. Nonlinear feature based classification of speech under stress. IEEE Trans. Audio Speech Language Process. 9, 201–216.

1275

1276

1277